Author's response to reviews

Title: Association of Biomass Fuel Use with Acute Respiratory Infections among Under-Five Children in a Slum Urban of Addis Ababa, Ethiopia

Authors:

Habtamu Sanbata Mr (habtish217@gmail.com)
Araya Asfaw PhD (arayaa@hoarec.org)
Abera Kumie MD, PHD (aberakumie2@yahoo.com)

Version: 5 Date: 30 August 2014

Author's response to reviews:


On behalf my colleagues I would like to thank you for the valuable comments and suggestions on the manuscript “Effects of Biomass Fuel Use on Acute Respiratory Illness among Children in Slum Urban of Addis Ababa City, Ethiopia”. Below please find our response to the comments.

1. The recommendation is accepted and we have edited Introduction part.

2. Method has been revised to include the comments.

3. Tables have been edited and presented in terms of cases

4. The paper has been edited carefully for language.

In general over all the manuscript has been revised. The paper presents the association between household fuel use and acute respiratory infection among children in Addis Ababa, Ethiopia. This study has significant contribution to reduce public health problem from household fuel use.

This study is not published or is not under consideration to be submitted to other journal

With best regards

Habtamu Sanbata

Corresponding author

School of Public and Environmental Health, Hawassa University

Email: habtish217@gmail.com, Habtamus@hu.edu.et
Cell Phone: 251913007924